Author’s response to reviews

Title: Rapidly evolving conjunctivitis due to Pasteurella multocida, occurring after direct inoculation with animal droplets in an immunocompromised host.

Authors:

Anthony Corchia (anthonycorchia@gmail.com)
Anne Limelette (alimelette@chu-reims.fr)
Béatrice Hubault (beatrice.hubault@gmail.com)
Ailsa Robbins (arobbins@chu-reims.fr)
Anne Quinquenel (aquinquenel@chu-reims.fr)
Firouze Bani-Sadr (fbanisadr@chu-reims.fr)
Yohan N’Guyen (yohan.nguyen@wanadoo.fr)

Version: 4
Date: 21 January 2015

Author’s response to reviews:

MS: 1480013852146397 R1

Responses to the editor:

The present revised manuscript has been substantially edited and we considered all the critiques raised by the editor.

1) Requesting for Copy-Edit

- Further consideration of your manuscript is conditional on improvement of the English used - please bear in mind that as we are a free-access publisher, we cannot bear the costs of copyediting English ourselves. Please ensure particular attention is paid to the abstract. You should have a native English speaking colleague help you with this, if possible, or you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

- Dear editor, this manuscript has been entirely reviewed and modified by Ian Robbins, PhD, born on January 1st 1957, in Redhill (Surrey). Please see the new version of the manuscript.

2) Please remove lines 1 to 4 on your title page (manuscript ID etc.)

- It has been done.